Author's response to reviews

Title: Sustained eradication of hepatitis C virus by low-dose long-term interferon therapy in a renal transplant recipient dual-infected with hepatitis B and C: a case report

Authors:

Ming-Ling Chang (mlchang8210@gmail.com)
Ping-Chin Lai (find94132@yahoo.com)
Chau-Ting Yeh (kitty8210@yahoo.com)

Version: 3 Date: 19 January 2011

Author's response to reviews:

Ms. Ref. No.: 2103739161421444

Title: Sustained eradication of hepatitis C virus by low-dose long-term interferon therapy in a renal transplant recipient dual-infected with hepatitis B and C: a case report

Dear Editors,

Thank you very much for your letter of Dec 21, 2010 with the opportunity to improve our manuscript. We have revised the manuscript according to the reviewers’ suggestions. The changes were marked with red letters in the revised manuscript. The followings are the point-by-point responses to the referee’s comments:

1. We had modified our paper according to the instruction for the author to conform to the journal style.

2. We had modified the dates (we erased the month) of the manuscript and figure to prevent jeopardize the anonymity of the patient

Thank you again for your help. We hope the paper is now acceptable.

Sincerely,

Chau-Ting Yeh, MD, PhD